Author’s response to reviews

Title: Study protocol for a randomized controlled trial comparing mindfulness-based cognitive therapy with maintenance anti-depressant treatment in the prevention of depressive relapse/recurrence: The PREVENT trial

Authors:

Wilm Kuyken (w.kuyken@exeter.ac.uk)
Sarah Byford (S.Byford@iop.kcl.ac.uk)
Richard Byng (richard.byng@pms.ac.uk)
Tim Dalgleish (tim.dalgleish@mrc-cbu.cam.ac.uk)
Glyn Lewis (Glyn.Lewis@bristol.ac.uk)
Rod Taylor (rod.taylor@pms.ac.uk)
Edward R Watkins (e.r.watkins@exeter.ac.uk)
Rachel Hayes (r.hayes@exeter.ac.uk)
Paul Lanham (judy@lanham.free-online.co.uk)
David Kessler (David.Kessler@bristol.ac.uk)
Nicola Morant (nm432@cam.ac.uk)
Alison Evans (alison.evans@exeter.ac.uk)

Version: 4 Date: 12 October 2010

Author’s response to reviews: see over
Dear Sir / Madam,

Re MS ID: 3115635564179691

I have responded to the requested changes as follows (changes in italics).

Formatting changes to be made:

Abstract: abstracts should not cite references, nor refer to figures or tables. Please remove the reference citations.

I have made these changes.

References: please provide a functioning link to the website in reference 15.

Completed.

References: please ensure that reference 26 is not a duplication of reference 29. If not, please provide full citation details for reference 26.

Completed. The in press version now replaced with the published version.

References: If possible, please provide a weblink for reference 9.

The citation numbers have changed, but I have provided a weblink for the Kings Report, citation number [5].

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

I have done so.

Yours faithfully,

Willem Kuyken, PhD